

Temi di discussione

(Working Papers)

Contagion in the CoCos market? A case study of two stress events

by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura





Temi di discussione

(Working Papers)

Contagion in the CoCos market? A case study of two stress events

by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura

Number 1201 - November 2018

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Emanuele Ciani, Nicola Curci, Davide Delle Monache, Francesco Franceschi, Andrea Linarello, Juho Taneli Makinen, Luca Metelli, Valentina Michelangeli, Mario Pietrunti, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print)

ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

CONTAGION IN THE COCOS MARKET? A CASE STUDY OF TWO STRESS EVENTS

by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura^{*}

Abstract

The post-crisis regulatory framework has fostered the development of the contingent convertible bonds (CoCos) market. These instruments permit banks to absorb losses as a going concern but their critics warn that they could have potentially destabilizing effects in stress situations. We analyse the dynamics of the European CoCos market during two stress episodes that occurred in 2016 and that were triggered by news of substantial unexpected losses faced by a European systemically important bank. Our econometric approach aims to disentangle the fundamental channel by which the contagion of such bank's distress spreads to the rest of the market from a possible CoCo-specific contagion channel. We find evidence of significant CoCo-specific contagion in the two stress episodes, which could be the result of investors' reassessment of the CoCos' riskiness or of uncertainty about their supervisory treatment. Moreover, we find that the CoCo-specific contagion was weaker in the second stress event, suggesting that as investors learn about the specific features of these instruments and their supervisory treatment the CoCos market becomes more resilient.

JEL Classification: G14, G21, G28. **Keywords**: CoCos, Basel III, contagion.

Contents

1. Introduction	5
2. Related literature	7
3. The CoCos market and the two Deutsche Bank stress events	
4. The transmission channels: fundamental and CoCo-specific contagion	13
5. Data	14
6. Empirical analysis	17
7. Further analysis and robustness	
8. Conclusions and policy implications	30
References	

^{*} Banca d'Italia, Financial Stability Directorate.

1. Introduction¹

In the aftermath of the global financial crisis bank capital requirements have been significantly increased to enhance the resilience of the system. In a number of jurisdictions the new regulatory framework allows banks to satisfy part of the additional capital needs with contingent convertible bonds (CoCos). The CoCos are hybrid debt instruments that in adverse contingencies can be used to recapitalize the banks by writing down their principal or converting it into equity.² Such a privileged regulatory treatment has spurred the growth of the CoCos market, with 208 billion USD of global issuances between the first issuance in 2009 and September 2014 (Avdjiev et al., 2017).

The rationale for the introduction of CoCo bonds is related to their contingent debt-equity nature in two ways. First, they provide an equity injection when bank capital is needed and the bank is still in going concern, i.e. when the bank is close to a given (regulatory) threshold but still above the regulatory minima. Second, the debt feature, that prevails except in rare distress episodes, makes them appealing to both banks, as they carry a lower cost relative to equity, and fixed income investors. Defendants of CoCos argue that these two properties render these instruments optimal to increase the loss absorption capacity of banks without imposing an excessive burden on bank profitability that could in turn negatively affect lending to the real economy (Duffie, 2009; Squam Lake Working Group, 2009; Flannery, 2016; Vallee, 2015). CoCos may also be beneficial by inducing timely recapitalisation to avoid costly dilution of profits or loss of control (Pennacchi, 2011; Calomiris and Herring, 2013). However, these arguments are by no means shared by all academics and practitioners. Sceptics in fact argue that the double nature of CoCos introduces complexity and uncertainty that, should a CoCo be on the eve of conversion, may prove destabilizing for the issuer bank and could spillover to the rest of the market (Goodhart, 2010; Admati et al., 2013; Allen, 2012).

In this paper we shed light on the potential destabilising effect of these securities by focusing on two stress periods experienced by Deutsche Bank (DB) during 2016, that created significant concerns on the value of its 4.6 billion EUR CoCo issuances. The two stress periods, that started in January and September 2016 respectively and lasted for a few weeks each, were initially triggered by announcements of substantial legal costs to be faced by DB. During the two events the yield-to-maturity (YTM) of DB's senior bonds and CoCos peaked to historical maxima. The unfolding of these events was accompanied by great concerns from practitioners and commentators about DB's CoCos, and in particular about both the possibility that their coupons would not be paid and that the supervisory authority could mandate their conversion.

¹Pierluigi would like to dedicate this work to Giovanni Abbate, a true mentor. The paper has benefited from a number of useful discussions with Piergiorgio Alessandri, Antonio Bassanetti, Nicola Branzoli, Luisa Carpinelli, Francesco Corsello, Fabio Panetta, Tommaso Perez, Enrico Sette, and Luigi Federico Signorini. The views expressed are those of the authors and should not be attributed to the Bank of Italy or the Eurosystem.

²Avdjiev et al. (2013) provide a description of the main design elements of CoCos.

The CoCos issued by the most important European banks also experienced severe tensions, especially during the January episode when their average YTM attained historical maxima. In that event, the YTM of the senior bonds issued by the same European banks also increased, although the level remained relatively contained. These dynamics could be the result of the contagion from a systemically important institution such as DB to other banks through "fundamental channels" stemming from, e.g., the network of interbank exposures or the possibility of fire-sales. Yet, the "overreaction" of the CoCo market relative to the behaviour of the senior bond market suggests that, in addition to fundamental distress propagation, the CoCo market could have been further destabilized by a broad reassessment of the riskiness of these instruments, ignited by the specific problems of DB's CoCos. Such a CoCo-specific contagion is precisely the type of concern raised by sceptics on these instruments and was frequently emphasized by commentators of these events. Our analysis of the two DB stress periods attempts at identifying whether or not there has been a CoCo-specific contagion beyond any fundamental distress propagation. By doing so, we investigate the potential unintended financial stability effects of a regulatory regime that encourages banks to issue CoCos.

In our empirical analysis we regress the daily YTM of the CoCos of European banks (excluding DB) on i) the institutions' stock return and the YTM of their senior bonds; and on ii) the YTM of DB's CoCos and its interaction with a dummy for each of the two events. The first set of controls aims at capturing the extent to which the CoCos YTM can be explained by the bank fundamental variables such as its solvency risk (proxied by the YTM of the bank's senior bonds) and its expected profitability (proxied by the bank's equity return). These variables allow us to control for the fundamental contagion from DB during the two distress episodes, and our specification thus captures how such contagion may have affected also CoCo prices. The second set of controls aims at capturing whether any inter-dependence between the Cocos of DB and those of other banks increased due to a Coco-specific contagion when DB was under distress.

Our results show strong evidence across all specifications on the existence of a CoCo-specific contagion channel in the propagation of the distress of DB to the rest of the European banks in the two episodes, which supports CoCos sceptics' claim that these instruments might be a source of financial instability when negative shocks occur. Moreover, we find that the behaviour of the CoCos issued by riskier banks does not significantly differ from those issued by safer institutions, which further suggests the existence of non-fundamental factors affecting the CoCo market dynamics in the two episodes. Yet, when comparing the CoCo-specific contagion between the two DB's events, we find that its magnitude was significantly weaker during the September episode. Such enhanced market stability could be consistent with investors' learning on the specificities of CoCos along time. The clarification by the European Banking Authority (European Banking Authority, 2016b) and the European Central Bank (European Central Bank, 2016a) on the supervisory treatment of these instruments, in particular with regard to the conditions that may call for the suspension of coupon payments, is also likely to have played a role in stabilizing the CoCos market

during the second stress period of DB. The weakening of a CoCo-specific contagion mechanism suggests that the potential destabilizing role of these securities – claimed by CoCos sceptics – might be only transitory and/or could have been mitigated by regulators' intervention, but we are probably not yet able to say the last word. Authorities should hence keep on monitoring the development of this market and its dynamics in situations of distress.

The rest of the paper proceeds as follows. Section 2 briefly recalls the literature related to this paper. Section 3 describes the regulatory treatment of CoCos, the market development, and the two DB distress events. Section 4 discusses the two transmission channels of contagion: a fundamental and a CoCo-specific one. Section 5 describes the data. Sections 6 present the empirical analysis. Section 7 conducts some further analysis and robustness exercises. Section 8 provides the policy implications of our findings and concludes.

2. Related Literature

Our paper belongs to a growing literature on CoCos starting with Flannery (2016) and Flannery (2005). The theoretical contributions initially dealt with how design features affect the valuation of these instruments (e.g. Posner, 2010; Pennacchi, 2011; Glasserman and Nouri, 2012). The possibility of multiple pricing equilibria for CoCos with market trigger and the ensuing manipulation incentives by the different stakeholders was uncovered in Sundaresan and Wang (2015), and subsequent contributions have identified design features that eliminate such pricing multiplicity (Calomiris and Herring, 2013; Pennacchi et al., 2014). Another strand of the theoretical literature has focused on how wealth transfers between the bank's stakeholders upon CoCo conversion may affect the ex-ante risk-taking decisions of the bank in a counter-productive way (Flannery, 2016; Hilscher and Raviv, 2014; Martynova and Perotti, 2018). Closer to the focus of our study, concerns that CoCos may not offer effective loss-absorption in times of distress have been raised by practitioners, academics and regulators (Pazarbasioglu et al., 2011; Delivorias, 2016).

The empirical literature on CoCos is scarce and has not addressed potential contagion problems associated with the conversion of these instruments. Berg and Kaserer (2015) investigate the impact of CoCos' issuances on banks' risk-taking using a sample of CoCo bonds issued by European banks over the period 2009-2013. The paper finds that some key-design features of CoCos, such as a too high conversion price lead the issuer bank to increase risk-taking. The literature has found mixed evidence on the effect of CoCos issuance on the riskiness of the issuer (Avdjiev et al., 2017; Goncharenko et al., 2017).

This paper is related also to the literature on contagion. The theoretical and empirical debate on what contagion means is far from conclusive, as emphasized by Pericoli and Sbracia (2003) and Forbes (2012). We interpret an increase in the comovement of CoCo prices after controlling for variations in the banks' fundamentals as evidence of CoCo-specific contagion. The notion of contagion as a transmission of shocks in excess of what can be explained by fundamentals is discussed in Claessens et al. (2001) and Forbes (2012), and has been adopted in the analysis by e.g., Calvo and Mendoza (2000), Bekaert et al. (2005) and Bekaert et al. (2014).

3. The CoCos market and the two Deutsche Bank stress events

3.1. CoCos market developments and regulatory treatment

The issuance of CoCos by European banks has grown steadily since the adoption of Basel III and its introduction in Europe by means of the Capital Requirements Directive IV (CRD IV) and the Capital Requirements Regulation (CRR).

Under Basel III, hybrid capital instruments, such as CoCos, are designed to absorb losses while the bank is still a going concern and are classified as Additional Tier 1 (AT1) capital. To qualify as AT1, CoCos must have, like common equity, a perpetual maturity and non-cumulative coupons, payable at the discretion of the issuer. They also need a loss-absorption trigger – activated when the CET1 capital ratio of the bank falls below a certain level – which allows the principal to be written down or converted to common equity without this event constituting a default of the issuer (Basel Committee for Banking Supervision, 2011). These features should enable to strengthen banks' capital position at a time when raising equity would otherwise be difficult.³ Furthermore, allowing the issuer to miss coupon payments can reduce pressure on liquidity. Finally, in a resolution process, CoCos are senior only to common equity.

In the EU, the CRD IV and the CRR set the criteria for CoCos to qualify as AT1 capital. In addition to the above mentioned harmonized conditions, EU regulation requires i) the loss-absorption trigger in terms of CET1 not to be lower than 5.125 per cent of risk-weighted assets; and ii) the coupon distribution on CoCos to be conditional on the banks' meeting the Combined Buffer Requirement (CBR).⁴ Should a bank not meet its CBR the amount of pay outs it can make in the form of dividends and coupon payments on AT1 instruments would be limited by the institution's Maximum Distributable Amount (MDA, CRD IV, Art. 141).

The favorable regulatory treatment of CoCos – combined with banks' need to strengthen their capital position during difficult market conditions for equity

³In addition to the automatic triggers, supervisory authorities can mandate the conversion or the write down of CoCos by discretionally activating the point of non-viability (PONV) triggers, if they believe that such action is necessary to prevent the bank from becoming insolvent (Delivorias, 2016).

⁴According to Art. 128 of the CRDIV the CBR is the sum of the Capital conservation buffer requirement and, when applicable, the Countercyclical capital buffer requirement, the G-SII and O-SII buffer requirements, and the Systemic risk buffer requirement.

issuances⁵ –, has spurred the development of the CoCos market in Europe. Based on Bank of America Merrill Lynch Contingent Capital Index (BAMLCCI), European banks have almost tripled their CoCo issuances, with the outstanding volume rising from 33.8 billion EUR to 107.4 billion EUR between January 2014 and November 2016 (Figure 1a).

The regulatory incentive in the growth of CoCos is confirmed by the fact that the vast majority of issuances qualify as AT1 capital, with only a small fraction of the total eligible as Tier 2 capital (Figure 1b). It is also worth highlighting that the Tier 2 CoCos have mostly been issued in the early days of the new-born market well before the approval of Basel III.⁶

Figure 1: The market of Contingent Convertible Bonds (billion EUR)

(b) European banks CoCos outstanding

by country and Basel III capital category

(November 2016)

(a) European banks CoCos outstanding issuances (January 2014 - November 2016)



Despite this strong growth, towards the end of 2015 uncertainties emerged in the interpretation of the legal framework for the computation of the MDA trigger, i.e. the level of the capital requirements below which a bank would need to calculate the MDA and restrict its pay outs, including those on AT1 instruments. In particular, it was unclear whether the CBR would come in addition of the 8 % Pillar 1 minimum requirements only, or in addition of both the Pillar 1 and Pillar 2 requirements. To clarify this issue, in December 2015 the European Banking Authority (European Banking Authority, 2015) published an opinion in which it reaffirmed that both Pillar 1 and Pillar 2 requirements should be below the CBR when determining the MDA trigger. It was also recognized that, given the importance of the MDA trigger to investors in AT1 instruments, there should be full disclosure of the Pillar 2 requirements. In early 2016, the European Central Bank (2016b) endorsed the

⁵This has occurred in particular in the preparatory phase of the Single Supervisory Mechanism, with the Asset Quality Review completed in October 2014, and with its entry into force.

⁶The first CoCo was issued by Lloyds Banking Group in November 2009.

EBA opinion. Notwithstanding such a clarification, uncertainties yet remained, this time due to the possibility that some banks could have breached the MDA trigger not because of actual losses but rather because of potential losses arising from the adverse scenario of the supervisory stress test, that would have been carried out over the following months.

Issuances of CoCos abruptly halted in early 2016, contemporaneously to the first DB stress period analyzed in this paper.

3.2. The two Deutsche Bank stress events

In this section we discuss in more detail each of the two distress episodes experienced by DB in 2016.

Event 1

The first event took place at the beginning of 2016 after the announcement of negative earnings expectations for 2015 which came as a surprise for market participants.

On January 20, 2016 DB announced that it expected an extraordinary high net loss of about 6.7 billion EUR for 2015, due to unanticipated very high litigation charges and restructuring costs and to cyclical market conditions (which led in particular to a fall in revenues from the bank securities trading unit). When the news was published, DB's debt securities reacted negatively (Figures 2a and 2b). The situation worsened a few days later on January 28 when the CEO of DB, Mr. John Cryan, addressed a message to the bank's employees. In particular, he clarified that the yearly losses for 2015 (revised by then to 6.8 billion EUR) had been driven mainly by regulatory and litigation provisions (5.2 billion EUR) and impairments on goodwill and other intangible assets (5.8 billion EUR) while, differently from the previous announcement, little emphasis was given to the market conditions. The news about the loss, coupled with the uncertainty about the bank's capital position relative to the MDA trigger⁷, raised dramatically the concerns on the capability of DB to meet the forthcoming payments on its CoCos (Glover, 2016). This in turn led the bank to take a number of initiatives to restore market confidence: on February 8, 2016, in a press release and in a message to the bank employees by the CFO of DB, Mr. Marcus Schenck, it was reaffirmed the bank cash availability to make the upcoming payments falling due on the CoCos.⁸ Then on the following day another message to the employees was sent by the CEO who claimed that "Deutsche Bank remains absolutely rock-solid".⁹ On February 12, 2016 the CFO announced a tender offer to

⁷The first CoCo was issued by Lloyds Banking Group in November 2009.

 $[\]label{eq:source} {}^{8}www.db.com/newsroom_news/2016/ir/deutsche-bank-publishes-updated-information-about-at1-payment-capacity-en-11391.htm\ {\rm and}\ www.db.com/newsroom_news/2016/ghp/a-message-from-marcus-schenck-deutsche-bank-s-additional-tier-1-at1-capital-en-11386.htm}$

 $^{^9}www.db.com/newsroom_news/2016/ghp/a-message-from-john-cryan-to-deutsche-bank-employees-0902-en-11392.htm$

buy back in the market some of the bank senior unsecured debt "taking advantage of market conditions to repurchase this debt, lowering its debt burden at attractive prices" and with "no impact on the bank's capacity to service coupons on its AT1 capital".¹⁰ The buy-back operation was clearly aimed at proving the soundness of the bank to the markets. On February 29, 2016, the CFO announced the success of the operation, which was welcomed by the market.¹¹ Since then conditions began to improve. The stabilization of the CoCos market might have also benefited from the recognition by the ECB, in February 2016, of the need to reduce uncertainty on the supervisory treatment of these instruments, and, in particular, on the conditions that would lead to the suspension of coupon payments. After long debate, the ECB announced on July 29, 2016 changes in the methodology to calculate the banks' MDA trigger that would effectively reduce the likelihood that banks would not be allowed to make CoCo coupon payments (see Section 8 for more details).

Before the described event the price of DB's CoCos were essentially stable. However, at the beginning of 2016 the YTM started rising with a sudden spike of almost 200 basis points (Figure 2a). In particular, the YTM jumped from an average of around 660-670 basis points in mid-January 2016, to more than 850 basis points in mid-February. We set the start date of the first distress episode on January 28 when, after the profitability announcement and the first message by the CEO of DB, the YTM of the DB's CoCos spiked by more than 100 basis points. We set the end date on February 29 when, after the announcement of the successful completion of a debt buy back, the YTM substantially declined.¹²

Event 2

The second distress event of DB started on September 12, 2016 when a number of articles in the press announced that the bank had received a 14 billion USD fine by the U.S. Justice Department. The news came after a series of other bad news that involved the bank over the summer,¹³ putting the institution under serious investors' scrutiny that was further exacerbated by the reports that the German government wouldn't help the ailing bank (Donahue, P., 2016). Both DB's CoCos and senior bonds yields took a hit, with the senior bonds almost reaching the pick observed in February 2016 (Figures 2a and 2b).

 $^{{}^{10}}www.db.com/newsroom_news/2016/ghp/a-message-from-marcus-schenck-deutsche-bank-announces-a-public-tender-offer-to-buy-back-debt-en-11399.htm$

 $^{{}^{11}}www.db.com/newsroom_news/2016/medien/deutsche - bank - announces - results - of - public-tender-offer-for-us-dollar-bonds-as-of-early-tender-date-en-11494.htm \\ {}^{12}\text{The results presented in the following sections are robust to setting the beginning of the distress}$

event to January 20, i.e. when DB announced for the first time the negative results. ¹³The most relevant are the following: i) in June the IMF stated that DB "appears to be the most

Fund, 2016); ii) also in June the US subsidiary of DB failed for the second consecutive year the stress test performed by the Federal Reserve, as the regulator rejected the bank capital plan (Federal Reserve Board, 2016); iii) in July the rating agency Standard&Poor's lowered its outlook on the bank's rating; iv) again in July the bank passed the EU-wide stress test by a tight margin (European Banking Authority, 2016a); and v) in August the bank was removed from the Stoxx 50 index.

Figure 2: The CoCos and senior bond markets during Deutsche Bank's CoCos distress (yield-to-maturity; January 2, 2015=100)



The rise in DB senior bonds and CoCos YTM started therefore on September 12, with an average increase on all issues by about 10 basis points in one day for both types of securities. Then other large one-day increases followed, particularly on September 16 when the news of the fine was confirmed – with the average YTM of DB senior bonds and CoCos increasing by 36 and 49 basis points respectively (Strasburg, J. (2016) and Davis, P.J. (2016)) – and on September 26 when the rumours about the government unwillingness to help the bank spread (Donahue, P., 2016) – with the average YTM of DB senior bonds and CoCos increasing by 22 and 29 basis points respectively. A number of interventions at the end of the September contributed to the recovery from this distress. In particular, on September 30 the CEO of DB reassured the market and the bank employees about the soundness of the bank's financial conditions;¹⁴ on the same day the Eurogroup President Mr. Jeroen Dijsselbloem stated that the 14 billion USD fine to DB was excessive,¹⁵ and unconfirmed reports claimed that the bank was about to agree on a reduction of the fine with the U.S. Department of Justice.¹⁶ Taking this information into account, in our empirical analysis we set September 12 as the start date of the second distress event, and September 30 as the end date.

 $[\]label{eq:source} {}^{14}www.db.com/newsroom_news/2016/media-reports-on-deutsche-bank-a-message-to-employees-from-john-cryan-en-11702.htm}$

 $^{^{15}} euobserver.com/tickers/135321$

¹⁶The rumors were indeed confirmed by the facts, as an agreement had been found in December when the fine was almost halved to 7.2 billion USD.

4. The transmission channels: fundamental and CoCo-specific contagion

The sequence of news in January and September 2016 led to mounting concerns on the ability of DB to pay the coupons on its CoCos or even of a possible write-down of their notional value. This information also increased the insolvency risk of DB, as shown by the large increases in the YTM of its senior debt (and by the negative stock returns). As exhibited in Figures 3 and 4, the rest of the European banking sector experienced also some distress: the YTM of the senior bonds and of the CoCos of European banks increased (and their stock prices fell). Such a co-movement may be the result of a propagation mechanism based on two contagion channels: a *fundamental* and a *CoCo-specific* one.

First, DB is a global systemically important institution whose financial difficulties may spillover to other institutions. This could result from the losses that DB's counterparties would suffer directly in case of insolvency of the institution, and indirectly from the exposure to other institutions through a loss cascade propagated by the network of interbank exposures. The distress of a systemic institution with important trading activities such as DB may also lead to substantial fire sales that would depress asset values and negatively affect the capitalization of banks with similar assets in their trading portfolio. Negative spillovers on the economy could also arise due to second round general equilibrium effects following the failure of a systemic institution. For example, it could reduce or make more costly the access to credit for banks' customers, that may as a response cut down investment or hiring; this in turn could depress the rest of the economy and, through this channel, the quality of other banks' assets. We refer to all these propagation mechanisms as "fundamental". They may explain all or part of the price evolution of banks' liabilities, including their CoCo instruments.

Second, both in January and September 2016, the news that DB may face substantial losses due to fines and legal charges immediately brought the attention to the institutions' capability to satisfy the coupon payments of its CoCos and to the possibility that supervisors would prohibit the bank from paying their coupons. More generally these events led to statements by practitioners and commentators alike on a broad reassessment of the riskiness of CoCos, and to concerns that their design and their debt-like features during normal times could have led investors to misunderstand these instruments. During these stress episodes, investors' uncertainty about CoCos could have led to adverse price dynamics extending from DB's CoCos to the CoCos issued by other institutions, well beyond what would be explained by fundamental factors (e.g. banks' credit risk, profitability, etc...). We refer to such a propagation as "CoCo-specific contagion".

The European banking sector was affected by the difficulties of DB in the two stress episodes, albeit to a different extent. While the severity of the distress of DB, as proxied by the change in the YTM of its senior bonds, was of a similar intensity in the two periods, the magnitude of the movements in the other European banks' senior bonds and CoCos YTM was significantly smaller in the second event. In the first event, the yield of the senior bonds of the major European banks experienced a short-lived increase, within the range of variation observed in the previous 12 months. This suggests a – possibly weak – fundamental contagion from DB to the other major European banks. In contrast, the European banks' CoCos yield hit an all-time high, which constitutes suggestive evidence of a CoCo-specific contagion.

In the second event, instead, the senior bond yield of the European banks remained essentially unchanged, which we can interpret as evidence of a very weak or even lack of fundamental contagion. The CoCos yields nevertheless significantly increased (although their maximum level remained below that attained in the first event), which suggests again the possibility of CoCo-specific contagion. The lower magnitude of the movements in the CoCos market during the second stress episode relative to the first could be the result of a lower fundamental contagion and/or a lower CoCo-specific contagion. A less intense contagion through the Coco-specific channel could be due to an ameliorated understanding by investors of the mechanisms and risks underlying these instruments following the January episode. Furthermore, the actions taken by the ECB during the summer, namely the reduction of the capital ratio below which coupon payments would not be allowed by regulators, could have contributed as well to mitigate volatility transmission through this channel.

In the next sections we investigate whether the described anecdotal evidence on both a fundamental and a CoCo-specific contagion, as well as their relative magnitude across the two events, can be confirmed by the empirical analysis.

5. Data

The main dataset used is provided by BAMLCCI. This index tracks on a daily basis the performance of contingent capital bonds publicly issued in the major markets. This dataset has been augmented with additional data at issuer level. In particular, for each bank we have selected all senior unsecured bonds from the Merril Lynch index ER00 and the daily (closing) data on equity prices from Bloomberg.¹⁷

For our purposes, we have selected data on securities issued by European banks between January 2014 and November 2016. Overall we have a sample of 102 CoCos issued by 22 banks located in 10 European countries (Table 1).¹⁸

¹⁷This matching procedure has led to a reduction of the original panel of banks from the BAMLCCI index.

¹⁸Information on issuers is aggregated at group level if the holding bank is represented in our sample. The banks included in our sample are the following: Erste Group (Austria), KBC Bank (Belgio), SocietÃI Generale, Credit Agricole and BNP Paribas (France), Aareal Bank and Deutsche Bank (Germany), Bank of Ireland (Ireland), Intesa SanPaolo and Unicredit (Italy), ABN Amro and Ing Group (the Netherlands), Banco Bilbao Vizacaya Argentaria, Bankinter, and Banco Santander (Spain), Credit Suisse and UBS (Swizterland), Barclays Plc., HSBC, Lloyds Banking Group, Royal Bank of Scotland, Standard Chartered (United Kingdom).

	Number of daily observations	Number of Issues	Number of issuers	Volume outstanding (billion EUR, November 2016)
Austria	105	1	1	0.5
Belgium	1,432	2	1	2.1
France	7,538	15	3	15.9
Germany	2,945	5	2	5.0
Ireland	364	1	1	0.8
Italy	3,096	6	2	4.5
The Netherlands	1,110	3	2	3.1
Spain	4,421	9	3	7.5
Switzerland	9,009	16	2	20.6
United Kingdom	16,571	44	5	35.9
Total	46,591	102	22	95.9

Table 1: Summary Statistics of Cocos Issuances

Table 2: Summary Statistics of Cocos Issuances by Loss Absorption Mechanism

	Number of issues			Volume o	utstanding (billion	EUR, November 20	016)	
	Temporary write-down	Permanent write-down	Equity conversion	Total	Temporary write-down	Permanent write-down	Equity conversion	Total
Austria	1	-	-	1	0.5	-	-	0.5
Belgium	1	1	-	2	1.4	0.7	-	2.1
France	14	1	-	15	15.2	0.7	-	15.9
Germany	5	-	-	5	5.0	-	-	5.0
Ireland	1	-	-	1	0.8	-	-	0.8
Italy	6	-	-	6	4.5	-	-	4.5
the Netherlands	1	-	2	3	1.0	-	2.1	3.1
Spain	-	1	8	9	-	1.0	6.4	7.5
Switzerland	-	15	1	16	-	20.6	-	20.6
United Kingdom	-	2	42	44	-	3.1	32.8	35.9
Total	29	20	53	102	28.4	26.2	41.4	95.9

The UK and the Swiss banking systems are the most represented in our sample, with both the largest number of issuances (44 and 16 issues respectively) and the highest volume outstanding (about 36 and 21 billion EUR respectively as of November 2016). Securities in the dataset are denominated in EUR, USD and GBP. In addition to daily closing market data (e.g. price, YTM, spread, and duration), information from BAMLCCI includes a large number of variables describing the main securities features (e.g. maturity date, rating, amount outstanding, issuance currency and whether the instrument is classified as AT1 or T2 capital). Additional information on CoCo instruments is obtained from Bloomberg. In particular, for each bond we have collected the main contractual characteristics, such as the loss absorption mechanisms (principal write-down/equity conversion), the type of write-down (temporary, permanent), the underlying trigger variable (e.g. Tier1 ratio, CET1) and the trigger level. Tables 2, 3, and 4 present more detailed summary statistics about this additional information for the selected sample of CoCos.

Statistics by country of the average daily YTM of the CoCos and the senior bonds as well as the average returns of equity prices are provided in Table 5.

Number of issues					Volume outstanding (billion EUR, November 2016)			
	CET1 ratio	Core Tier1 ratio	Total risk-based capital ratio	Total	CET1 ratio	Core Tier1 ratio	Total risk-based capital ratio	Total
Austria	1	-	-	1	0.5	-	-	0.5
Belgium	2	-	-	2	2.1	-	-	2.1
France	15	-	-	15	15.9	-	-	15.9
Germany	5	-	-	5	5.0	-	-	5.0
Ireland	1	-	-	1	0.8	-	-	0.8
Italy	4	-	2	6	4.0	-	0.5	4.5
the Netherlands	3	-	-	3	3.1	-	-	3.1
Spain	9	-	-	9	7.5	-	-	7.5
Switzerland	16	-	-	16	20.6	-	-	20.6
United Kingdom	25	19	-	44	35.9	-	-	35.9
Total	81	19	2	102	95.4	0.0	0.5	95.9

Table 3: Summary Statistics of Cocos Issuances by Trigger Variable

 Table 4: Summary Statistics of Cocos Issuances by Trigger Level

	Number of issues					
	Low-trigger			High-trigger	Total	
	5%	$5,\!125\%$	6%	7%		
Austria	-	1	-	_	1	
Belgium	-	1	-	1	2	
France	-	14	-	1	15	
Germany	-	4	-	1	5	
Ireland	-	1	-	-	1	
Italy	-	4	2	-	6	
the Netherlands	-	1	-	2	3	
Spain	-	8	-	1	9	
Switzerland	7	4	-	5	16	
United Kingdom	19	-	-	25	44	
Total	26	38	2	36	102	

 Table 5: Summary Statistics of Cocos Issuances by Trigger Level

	CoCo (aver basis	o b age points	oonds ytm,)	Senic (aver basis	or age points	bonds ytm, s)	Stock retur point	ks (av rn, ts)	erage basis
	2014	2015	2016	2014	2015	2016	2014	2015	2016
Austria	-	-	853	-	-	137	-	-	0.27
Belgium	611	595	601	72	104	105	0.04	0.09	0.01
France	682	665	685	170	135	92	-0.04	0.03	0.02
Germany	615	652	742	167	179	219	-0.13	-0.04	-0.12
Ireland	-	711	739	-	107	61	-	-0.05	-0.21
Italy	720	711	771	187	117	94	-0.01	0.01	-0.31
the Netherlands	-	621	610	-	111	87	-	-0.07	0.04
Spain	720	693	746	114	113	82	-0.03	-0.14	-0.03
Switzerland	528	535	550	92	62	61	-0.06	0.02	-0.12
United Kingdom	649	610	664	187	145	117	-0.02	-0.05	-0.01

6. Empirical analysis

6.1. Set-up

In order to disentangle fundamental from CoCo-specific contagion, we define the following baseline model:

$$Y_{i,j,t} = \alpha_{i,j} + \beta Y_{i,j,t-1} + \sum_{k=1,2} \gamma_k X_{k,j,t} + \delta Y_{DB,t} + \sum_{s=1,2} \varphi_s Y_{DB,t} Dummy_{s,t} + \theta Z_{i,j,t} + \varepsilon_{i,j,t}.$$
(1)

The dependent variable, $Y_{i,j,t}$, is the daily YTM of CoCo *i*, issued by a European bank *j* other than DB, at time *t*. We include as a regressor one lag of the dependent variable, $Y_{i,j,t-1}$, to account for serial correlation in yields. The pair of bank specific controls $X_{k,j,t}$, with k = 1, 2, consists of the daily average YTM of the senior bonds (k = 1) issued by bank *j* and of the daily stock return (k = 2) of bank *j*, both at time *t*. By using these controls we aim to capture the variation in the CoCos YTM that can be explained by the changes in bank fundamental conditions such as its solvency (proxied by the YTM of the bank's senior bonds) and its expected profitability (proxied by the bank's equity return). To the extent that there is a fundamental contagion from DB to the other banks' CoCos YTM during the two distress episodes, it should be reflected by these variables.

The variable $Y_{DB,t}$ is the daily average YTM of the CoCos issued by DB which we use to control for the possible co-movement between the CoCos of DB and those of the other banks. In particular, the coefficient of $Y_{DB,t}$ in the baseline regression would capture the average inter-dependence in the YTM of CoCos, along all the sample and time span considered, which could result from unobserved variables that affect the pricing of all CoCos, but not that of the other banks' securities. We also interact $Y_{DB,t}$ with the dummy variables $Dummy_{s,t}$, with s = 1, 2, which identify the two DB stress periods as previously defined. The coefficients of the interactions aim to capture whether there has been a change in the inter-dependence between the CoCo of DB and those of other banks when DB was under distress (and investors spotted the riskiness of its CoCos).

Finally, $Z_{i,j,t}$ is a vector of CoCo-specific variables that includes the type of action upon conversion (principal write-down or equity conversion), the type of write-down (temporary or permanent), the regulatory treatment of the instrument (AT1 or Tier 2), the trigger level (high or low¹⁹), the possibility for the supervisory authority to suspend coupon payments, and the size of the issue.

We use the baseline regression to test the two following hypothesis:

¹⁹We follow Avdjiev et al. (2013) and define a high (low) trigger CoCo bond as one having a trigger level above (below) 6 per cent of the ratio between Common Equity Tier 1 capital and Risk-weighted assets (CET1/RWA). The results of the analysis do not change to setting a threshold at 5.125 per cent, the regulatory minima for a CoCo to qualify as AT1.

Hypothesis 1: Existence of CoCo-specific contagion in the first and/or the second stress period

We say that there is evidence of CoCo-specific contagion in the stress period s if the coefficient of $Y_{DB,t}Dummy_{s,t}$ is statistically significant and positive. Our definition of contagion is consistent with that of Forbes and Rigobon (2002) that consider contagion to be an increase in cross-dependence across markets during crises periods. Note that since we control for the banks' equity and senior unsecured debt price, the potential fundamental contagion from DB to the other banks' CoCos should be transmitted through the effect of DB's distress on these variables. Our baseline pricing model does not capture the potential non-linearities that can be especially important when banks financial position is eroded and the pricing of their CoCos becomes more information sensitive. We conduct a robustness exercise in Section 6.2 to address this issue.

Hypothesis 2: Reduction in CoCo-specific contagion in the second stress period relative to the first one

Conditional on the existence of CoCo-specific contagion in both periods, we say that there is evidence of a reduction of contagion from the first to the second period if the interacted coefficient of $Y_{DB,t}Dummy_{2,t}$ is lower than that of $Y_{DB,t}Dummy_{1,t}$. Since the yields' volatility may change during the two stress periods, we look at the coefficients' economic significance which we compute by normalizing each coefficient by the standard deviation of the YTM of DB's CoCos observed in each corresponding stress period.

6.2. Estimation

Equation (1) is first estimated using a Generalized Least Square (GLS) estimator with random effects, to control for the issue-specific characteristics of the CoCos. We estimate the model by clustering the standard errors by bank and time and by allowing for intragroup correlation (i.e. relaxing the usual assumption that the observations are independent), since the null hypothesis of no serial correlation in the residuals is rejected by the Wald test discussed in Wooldridge (2002) and for which Drukker et al. (2003) shows the good size and power properties.

In addition, as the Likelihood Ratio test shows that there is heteroscedasticity across groups, we also run a Feasible GLS (FGLS) estimation with heteroskedastic error structure across panels and a panel-specific AR(1) process within panels.

Finally, although the time dimension of our panel is relatively large (746 days) and hence we could expect a priori the dynamic panel bias of GLS and FGLS estimations to be small, we check their results against the generalized method of moments (GMM) estimator proposed by Blundell and Bond (1998), also known as System GMM. The latter framework accounts for endogeneity, controls for unobserved heterogeneity, and deals with the biases and inconsistencies typical of least square estimations, provided that the model is not subject to serial correlation of order two and the instruments used are valid. We estimate the System GMM with a two-step estimator robust to panel-specific autocorrelation and heteroscedasticity.

6.3. Results

The results from the estimation of equation (1) are shown in Table 6, with the GLS estimation reported sub (i), the FGLS sub (ii), and the System GMM sub (iii). The three estimates confirm that the CoCo YTM is characterised by an AR(1) process, as shown by the very large and significant coefficient of the autoregressive term. The results also confirm, as expected, that the CoCo YTM is inversely related with the bank stock returns, as shown by the significance of the stock return variable, and that it positively co-moves with the senior bonds YTM. For the latter variable the evidence is however somewhat less robust due to its low significance, observed in two estimations only (GLS and GMM). These findings show that there is a close relationship between the banks' fundamental conditions and the pricing of their CoCos. Any effect of the two DB events on the other banks' CoCos resulting from the fundamental contagion channel previously described should therefore be captured by the significance of the senior bond and stock return variables. Table 6 shows also that there is limited evidence of co-movement between the CoCos of DB and those of the other banks over the entire time-span as the DB's CoCos variable is significant only under the FGLS estimation sub (ii). Such significance could be due to unobservable variables that affect all CoCos and that might bias our estimates. We analyse therefore the robustness of our results to the possibility of omitted variables in section 6.3.

As for the variables of direct interest to test the CoCo-specific contagion hypotheses described in the previous paragraph, we find that the two interacted variables $Y_{DB,t}Dummy_{s,t}$ are both positive and very significant across the three This confirms Hypothesis 1 about the existence of alternative estimations. CoCo-specific contagion during the two stress episodes. We also observe that the size of the interacted variable is larger in the first episode, which suggests a lower intensity of the CoCo-specific contagion in the second event. In order to confirm that this is indeed the case, we assess the economic significance of the coefficients and take into account the volatility observed during the two events, as a higher volatility in the second event could have been in principle the reason for the lower estimated coefficients. However, we find that this is not the case. After adjusting the coefficients of $Y_{DB,t}Dummy_{s,t}$ for the standard deviation of the $Y_{DB,t}$ during each event (89 and 65 basis points, respectively), we find that the relative magnitude of the coefficient for the first event is more than four times larger than that for the second event. We conclude therefore that Hypothesis 2 about a reduction of the CoCo-specific contagion between the first and the second distress period is validated by the data.

In terms of CoCo-specific characteristics, there is little evidence of their role in explaining the YTM of CoCos. All but one of the variables capturing the different features of the CoCos are in fact not significant in all three estimations. The variable controlling for the possibility of the regulator to cancel interest payments is significantly related to the CoCos YTM in sub (i), while the issue-specific subordination (AT1 vs Tier 2) is marginally significant in sub (iii). The negative sign of the coefficient of the former variable indicates that those CoCos that allow the discretionary suspension of interest payments by the regulator have lower YTM (higher prices), a somewhat counterintuitive result. The negative sign of the coefficient of the issue-specific subordination variable indicates that the more senior the CoCo the lower its YTM (the higher its price). Overall, this evidence suggests that in the pricing of a CoCo – undoubtedly a complex instrument to price – its contractual features do not seem to play a significant role.

	(1)	(0)	(0)
VARIABLES	$^{(1)}_{\text{GLS RE}}$	$^{(2)}$ FGLS	(3) System GMM
Lag CoCo ytm	0.996***	0.998***	0.995***
	(0.001)	(0.000)	(0.001)
Senior bond vtm	0.004**	0.001	0.009*
Sellior Sella y th	(0.002)	(0.001)	(0.005)
Stock roturn	0.726***	0.608***	1.074***
Stock letuin	(0.057)	(0.008)	(0.197)
	(0.001)	(0.012)	(0.157)
Coco DB	(0.001)	0.002^{++++}	(0.002)
	(0.001)	(0.001)	(0.002)
$Coco DB \ge event1$	0.113***	0.116***	0.146***
	(0.012)	(0.003)	(0.024)
$Coco DB \ge event2$	0.028^{***}	0.022^{***}	0.046^{***}
	(0.006)	(0.006)	(0.006)
Event1	-86.926***	-88.677***	-113.181***
	(9.420)	(2.436)	(18.174)
Event2	-20.557***	-16.328***	-33.541***
	(4.685)	(4.264)	(4.007)
Coase abarratoristics	(1.000)	()	(1.001)
Cocos characteristics			
Subordination type	-0.738	-0.156	-1.356^{**}
	(0.473)	(0.110)	(0.600)
Equity conversion	-0.237	-0.056	-0.343
1 0	(0.269)	(0.099)	(0.315)
Permanent write-down	0.035	-0.125	0.330
	(0.209)	(0.112)	(0.481)
Cancellation by regulators	0.179*	0.002	0.270
Cancenation by regulators	(0.093)	(0.052)	(0.244)
The investment of the second s	(0.055)	(0.012)	0.200
Ingger level	-0.210	-0.120	-0.209
	(0.190)	(0.065)	(0.295)
Size	0.160	-0.064	0.313
	(0.190)	(0.077)	(0.275)
Constant	1.163	-0.033	1.333
	(1.501)	(0.491)	(1.656)
Observations	30512	30512	30512
Hansen test	-	-	82.29
p-value			0.149
Arellano Bond test for $AR(1)$			-1.691
p-value			0.0909
Arellano Bond test for $AR(2)$			1.489
p-value			0.136
Number of instruments			85

Table 6: Fundamental and Coco-specific Dependence of Cocos Yield-to-matuirty During DB Events

Estimates of the yield to maturity of the Cocos issued by European banks, excluding DB. Random effect estimations is shown sub (1); Feasible GLS estimation with heteroskedastic error structure across panels and a panel-specific AR(1) process within panels is reported sub (2); Blundell-Bond system GMM estimates are shown sub (3). For the GMM, Hansen, AR(1), and AR(2) tests are provided. Robust standard errors are in parentheses. *** p < 0.01,** p < 0.05, * p < 0.1.

7. Further analysis and robustness

In the previous section, we have found compelling evidence on the existence of CoCo-specific contagion during the DB distress episodes that is not explained by the dynamics of banks' fundamental risk. In this section, we conduct some further analysis and robustness exercises to check the validity of our results.

7.1. Bank riskiness and CoCo dynamics

To further explore the interaction between fundamental and CoCo-specific contagion we augment the baseline model in equation (1) by the triple interaction $Y_{DB,t}Senior_{i,t}$ $Dummy_{s,t}$, i.e. between the average YTM of DB's CoCos, the average YTM of the bank j's senior unsecured bonds, and the event dummies. A positive coefficient on this triple interaction would mean a higher sensitivity of the price of the CoCos issued by the riskier banks to the developments of DB. Such type of behaviour would be the one to expect if the dynamics of CoCos were entirely driven by fundamentals, as the CoCos issued by riskier banks are more likely to suffer MDA restrictions on coupon payments or a potential conversion. In contrast, a non-significant coefficient would show that the interdependence between European banks' CoCos and those issued by DB does not depend on the riskiness of the issuer. This in turn would provide evidence of non-fundamental factors driving the co-movements in CoCos prices and would constitute additional evidence of CoCo-specific contagion. The results of the robustness analysis are presented in Table 7. The non-significance of the interactions in all but one case provides suggestive evidence on the disconnection between the CoCos market dynamics and the fundamental risk of the issuing banks and further points towards the existence of CoCo-specific contagion during the DB stress periods.

7.2. Non linearities in CoCo prices during stress events

As an additional robustness exercise, we test the validity of our results against the possible role of non-linearities between bank risk and the price of CoCos during the two DB distress periods. We do so by estimating equation (1) augmented by the interactions $Senior_{j,t}Dummy_{s,t}$ and $Stock_{j,t}Dummy_{s,t}$, that is, between the two dummy events and the senior bond YTM of bank j and the stock return of bank j, respectively. The results shown in Table 8 confirm the validity of our findings as the magnitude and significance of the coefficients of $Y_{DB,t}Dummy_{s,t}$ are essentially unchanged relative to their values in Table 6.

	(1)	(2)	(3)
VARIABLES	GLS RE	FGLS	System GMM
	0.000***	0.000****	0.00 (***
Lag CoCo ytm	0.996^{***}	(0.998^{***})	0.994^{***}
Contan handatur	0.001)	(0.000)	(0.001)
Senior bond ytm	(0.003)	(0.001)	(0.014)
Stock return	-0 727***	-0.609***	-0.614***
Stock lotali	(0.057)	(0.012)	(0.154)
Coco DB	0.001	0.002***	0.002
	(0.001)	(0.001)	(0.003)
Coco DB x event1	0.110***	0.114***	0.120***
	(0.013)	(0.003)	(0.029)
Coco DB x event2	0.028^{***}	0.022***	0.048^{***}
	(0.006)	(0.006)	(0.006)
Event1	-85.810***	-88.047***	-93.102***
	(9.425)	(2.442)	(21.941)
Event2	-20.814***	-16.465***	-36.246***
	(4.079)	(4.200)	(4.241)
Senior bond x event1 x CoCo DB	(0.000)	(0.000^{***})	(0.000)
Senior bond y event? y CoCo DB	0.000)	0.000	(0.000)
Semor bond x event2 x CoCo DB	(0.000)	(0.000)	(0.000)
Cocos characteristics	(0.000)	(0.000)	(0.000)
	0.791	0.150	1 409**
Subordination type	-0.731 (0.472)	-0.159 (0.110)	-1.493^{m}
Equity conversion	0.22	0.053	0.253
Equity conversion	(0.269)	(0.099)	(0.349)
Permanent write-down	0.028	-0.125	0.537
	(0.208)	(0.112)	(0.556)
Cancellation by regulators	-0.176*	-0.094	-0.290
	(0.093)	(0.072)	(0.303)
Trigger level	-0.211	-0.120	-0.042
	(0.196)	(0.083)	(0.307)
Size	0.145	-0.069	0.399
	(0.187)	(0.077)	(0.313)
Constant	1.430	0.120	0.475
	(1.488)	(0.493)	(1.873)
Observations	20519	20519	20519
Hansen test	00012	30312	82.55
p-value			0.145
Arellano Bond test for $AR(1)$			-1.652
p-value			0.0985
Arellano Bond test for $AR(2)$			1.373
p-value			0.170
Number of instruments			87

Table 7: Fundamental and Coco-specific Dependence of Cocos Ytm During DB Events: Bank Risk and Cocos YTM Robustness

Estimates of the yield to maturity of the Cocos issued by European banks, excluding DB. Random effect estimations is shown sub (1); Feasible GLS estimation with heteroskedastic error structure across panels and a panel-specific AR(1) process within panels is reported sub (2); Blundell-Bond system GMM estimates are shown sub (3). For the GMM, Hansen, AR(1), and AR(2) tests are provided. Robust standard errors are in parentheses. *** p < 0.01,** p < 0.05, * p < 0.1.

VARIABLES	(1) GLS RE	(2) FGLS	(3) System GMM
Lag CoCo vtm	0.996***	0.998***	0.994***
Lag coco jun	(0.001)	(0,000)	(0.002)
Conion hand atm	0.002*	0.001	0.014**
Senior bond ythi	(0.003)	(0.001)	(0.014)
	(0.002)	(0.001)	(0.000)
Stock return	-0.649***	-0.564***	-1.604***
	(0.064)	(0.013)	(0.243)
Coco DB	0.001	0.002***	0.005*
	(0.001)	(0.001)	(0.003)
Coco DB x event1	0.111^{***}	0.114^{***}	0.151^{***}
	(0.012)	(0.003)	(0.017)
Coco DB x event2	0.028^{***}	0.022^{***}	0.032^{***}
	(0.006)	(0.006)	(0.006)
Event1	-86.970***	-88.436***	-117.181***
	(9.475)	(2.412)	(12.384)
Event2	-21.154***	-16.727***	-24.092***
	(4.754)	(4.297)	(4.288)
Senior bond x event1	0.013	0.000***	0.002
Senior Bolid & events	(0.013)	(0.003)	(0.002)
Conion bond w event?	0.007	0.004	0.011
Senior bond x event2	(0.007)	(0.004)	(0.011)
	(0.000)	(0.000)	(0.010)
Stock return x event1	-0.529***	-0.374***	0.781***
	(0.180)	(0.030)	(0.159)
Stock return x event2	-0.169	-0.072	0.691***
	(0.171)	(0.114)	(0.241)
Cocos characteristics			
Subordination type	-0.730	-0.141	-1.441**
JI	(0.471)	(0.108)	(0.706)
Equity conversion	-0.218	-0.071	-0.320
Equity conversion	(0.268)	(0.100)	(0.431)
Permanent write down	0.022	0.127	0.480
i ermanent write-down	(0.208)	(0.137)	(0.533)
	0.179*	0.004	0.000
Cancenation by regulators	-0.178	(0.094)	-0.290
	(0.035)	(0.072)	(0.320)
Trigger level	-0.210	-0.124	-0.097
_	(0.196)	(0.082)	(0.393)
Size	0.138	-0.068	0.441
	(0.186)	(0.077)	(0.350)
~			
Constant	1.475	0.111	-1.286
	(1.497)	(0.491)	(2.746)
01	20512	90510	20512
Ubservations Hangon tost	30512	30512	30512
nansen test			04.90
Arellano Bond test for $\Delta R(1)$			-1 768
p-value			0.0770
Arellano Bond test for $AB(2)$			1.717
p-value			0.0859
Number of instruments			109

Table 8: Fundamental and Coco-specific Dependence of Cocos YTM During DB Events:Non-linearities in the Price of Bank Risk

Estimates of the yield to maturity of the Cocos issued by European banks, excluding DB. Random effect estimations is shown sub (1); Feasible GLS estimation with heteroskedastic error structure across panels and a panel-specific AR(1) process within panels is reported sub (2); Blundell-Bond system GMM estimates are shown sub (3). For the GMM, Hansen, AR(1), and AR(2) tests are provided. Robust standard errors are in parentheses. *** p < 0.01,** p < 0.05, * p < 0.1.

7.3. Omitted variables

In two out of the three estimations presented in Table 6 the DB CoCo variable is not a significant driver of other banks' CoCo YTM under normal market conditions, as we expected. However, in the estimation sub ii) it turned out to be significant. Even though it is so only in one out of three estimations we suspect that its significance might be due to a possible omitted variable problem, which is to say that omitted or unobservable variables might drive both DB's and other banks' CoCos YTM, hence the significance of the DB's CoCos in Table 6.

In order to test for this possibility we perform a four-steps procedure which aims to separate the effect of any omitted variable from any DB-specific effect. The procedure is the following. First, we run a principal component analysis (PCA) on the dataset of CoCos YTM, which includes both the CoCos issued by DB and those issued by the other banks. By means of the PCA, we aim to identify those components that capture the CoCos co-movements. As such, they should also reflect the effect of any unobserved variable that affect all CoCos similarly. Second, we select the components that together explain a significant proportion of the overall variance of our CoCos dataset; in particular, we consider the first three components that together account for about 84 per cent of the variance. Third, we use them as regressors to explain the DB's CoCos YTM; the resulting residuals are stored as they constitute the "purely" idiosyncratic part of the DB's CoCos YTM which remains unexplained from the regression. Fourth, with an approach similar in spirit to the forecasting method proposed by Stock and Watson (2002), we study the CoCo YTM running again the FGLS estimate of equation (1), but instead of including the DB CoCo YTM and its interactions with the two stress events, we use as regressors: i) the first three principal components selected in the previous step; and ii) the stored DB-specific residuals, which we also interact with the two DB stress events. Following this procedure we should be able to isolate any true DB-specific effect from the effect that may be due to any omitted or unobservable variable.

The results of the FGLS regression, which are comparable to those presented in Table 6 sub (ii), are shown in Table 9. They confirm our hypothesis that the significance in normal times of the DB CoCos evidenced in Table 6 sub (ii) is more likely to be due to an omitted variable problem rather than to a DB-specific role in explaining other banks' CoCos YTM. In fact, results show that the DB-specific variable (DB residuals) is not significant in explaining other banks' CoCos YTM in normal times, once we control for possible omitted variables with the principal components (significant in two cases). These results also strengthen our argument about the existence of a CoCo-specific contagion channel from DB to other banks during the two stress events, as the interactions of the DB residuals variable with the stress event dummies are both very significant.

	(1)
VARIABLES	FGLS
Lag CoCo vtm	0 997***
Lag COCO your	(0.000)
Senior bond ytm	0.002
v	(0.001)
Stock return	-0.731***
	(0.016)
Component 1	0.426
	(0.275)
Component 2	1.035*
	(0.579)
Component 3	0.298**
	(0.131)
DB residuals	-0.042
	(0.030)
Event1	-54.725***
T 10	(3.374)
Event2	-71.830^{***}
	(12.914)
DB residuals x event1	(0.074^{++++})
DP residuals y event?	0.005***
DD residuais x event 2	(0.095^{+++})
Cocos characteristics	(0.017)
Subordination type	0.251
Subordination type	(0.174)
Fauity conversion	0.040
Equity conversion	(0.159)
Permanent write-down	-0.188
	(0.171)
Cancellation by regulator	-0.122
	(0.111)
Trigger level	-0.172
00	(0.131)
Size	-0.151
	(0.118)
Constant	31 992
Constant	(21.310)
Observations	16719

Table 9: Controlling for Omitted Variables

Feasible GLS estimation of the yield to maturity of the Cocos issued by European banks, excluding DB. Robust standard errors in parentheses. The estimation allows for heteroskedastic error structure across panels and a panel-specific AR(1) process within panels. Robust standard errors are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.



Figure 3: Cross-sectional distribution of the estimated CoCo x DB interacted coefficients

7.4. Heterogeneity in CoCo-specific contagion

The results discussed so far describe the average features of our sample of CoCos issued by European banks. An interesting question is how "representative" these numbers are: does the CoCo-specific contagion during the DB events differ significantly and systematically across CoCo issues? The large-T nature of our panel allows us to answer this question by estimating issue-specific models and scrutinizing the cross-sectional distributions of the coefficients.

The results confirm the findings of the panel estimation. Figure 3 shows the distribution of the coefficients of $Y_{DB,t}Dummy_{s,t}$ obtained from OLS estimates of equation (1) performed at Isin level. The pooled estimates appear to be a good summary of the data as the banks' CoCo YTM are indeed positively correlated with the YTM of the DB CoCos both in the first and in the second event. Coherently with the panel estimates, the distribution of the estimated coefficients for the first turbulent period is upward shifted, albeit with a higher dispersion, relative to that for the second period, which is more centred around zero. The key message is therefore that our panel estimations provide a good description of most CoCo issues in the sample.

Table 10 confirms the validity of these results by showing the share of significant coefficients of $Y_{DB,t}Dummy_{s,t}$ across the OLS estimates of equation (1) performed at Isin level. The estimated coefficients (at either 1 or 5 percent level) are significant in a very high number of CoCo issues across the two events (74 and 84 per cent respectively) and for almost all issuers (90 per cent). Interestingly, in the second event the significance is also more concentrated at the 1 per cent level.

7.5. Banks' business model and CoCo-specific contagion

To further explore the nature of the CoCo-specific contagion in the two events, we consider the possible role of the banks' business model. In other words, we want to

Estimated significant coefficients (share of total estimates)						
Significance level	CoCo DB x event1	CoCo DB x event2				
1 percent	0.55	0.80				
5 percent	0.19	0.04				
Total	0.74	0.84				

Table 10: Significance of Coco-specific Contagion Across Issue-specific OLS Regressions

investigate whether banks with a business model closer to that of DB, i.e. investment banking, might have experienced higher contagion. Using information from banks' balance-sheet at the end of 2016, we consider a bank to be "Investment bank" (IB) if its trading assets account for more than 30 per cent of total assets. We then define a dummy variable IB_j which captures whether a bank j's business model is IB or not, and estimate again equation (1) augmented by IB_j and its interaction with $Y_{DB,t}Dummy_{s,t}$.

The results in Table 11 show that investment banks might face on average a higher CoCo YTM (sub (iii)). More importantly in the context of the analysis, they also indicate that being an investment bank, i.e. having a business model closer to that of DB, might have implied a more severe CoCo-specific contagion during the first DB distress event, given the significance of the coefficient of the triple interaction $IB_{i}Y_{DB,t}Dummy_{s,t}$ (sub ii)).

	(1)	(2)	(3)
VARIABLES	GLS RE	FGLS	System GMM
Lag CoCo ytm level coco	0.996***	0.998***	0.992***
0	(0.001)	(0.000)	(0.002)
Senior bond ytm	0.004^{*}	0.001	0.015**
	(0.002)	(0.001)	(0.006)
Stock return	-0.725***	-0.608***	-0.574***
	(0.057)	(0.012)	(0.146)
Coco DB	0.001	0.002***	0.003
(1 DD 11	(0.001)	(0.001)	(0.003)
Coco DB x event1	(0.113^{***})	(0.003)	(0.030)
Case DB w event?	0.012)	0.0000)	0.040***
Coco DB x event2	(0.028) (0.006)	(0.022)	(0.049)
Event1	-87 106***	-88 730***	-85 072***
1.101101	(9.382)	(2.435)	(22.728)
Event2	-20.625***	-16.327***	-35.698***
	(4.656)	(4.263)	(4.179)
IB	0.207	0.084	3.427***
	(0.161)	(0.082)	(1.243)
IB x event1 x CoCo DB	0.003	0.002***	0.001
	(0.002)	(0.000)	(0.002)
IB x event2 x CoCo DB	0.000	0.000	0.001 (0.001)
Cocos characteristics	(0.001)	(0.001)	(0.001)
Subordination type	-0.823*	-0.187*	-2.439*
	(0.492)	(0.112)	(1.391)
Equity conversion	-0.157	-0.026	0.659
	(0.264)	(0.101)	(0.746)
Permanent write-down	0.131	-0.072	1.822
	(0.236)	(0.117)	(1.170)
Cancellation by regulator	-0.251**	-0.118	-1.097
	(0.099)	(0.074)	(0.778)
Trigger level	-0.145 (0.183)	-0.108	(0.891)
C:	0.105	0.070	(0.715)
Size	(0.125)	(0.079)	(0.583)
	(0.100)	(0.011)	(0.000)
Constant	1 352	0.025	0.350
Constant	(1.496)	(0.491)	(2.350)
	()		
Observations	30512	30512	30512
Hansen test			81.90
Arollano Bond toot for AD(1)			0.177
nienano dono test ior AK(1)			-1.002
Arellano Bond test for AR(2)			1.376
p-value			0.169
Number of instruments			89

 Table 11: Fundamental and Coco-specific Dependence of Cocos YTM During DB Events: the Role of Bank Business Model

Estimates of the yield to maturity of the Cocos issued by European banks, excluding DB. Random effect estimations is shown sub (1); Feasible GLS estimation with heteroskedastic error structure across panels and a panel-specific AR(1) process within panels is reported sub (2); Blundell-Bond system GMM estimates are shown sub (3). For the GMM, Hansen, AR(1), and AR(2) tests are provided. Robust standard errors are in parentheses. *** p < 0.01,** p < 0.05, * p < 0.1.

8. Conclusions and policy implications

Our empirical analysis provides evidence of, on the one hand, the existence of a CoCo-specific contagion channel in the propagation of the distress of DB to the rest of the European banks in the two stress episodes considered, and on the other hand, the fact that such a contagion channel seems to have weakened in the second episode.

From a financial stability perspective, the widespread turbulence in the CoCos market in front of idiosyncratic shocks not totally explained by fundamentals supports the idea of CoCo sceptics that these instruments might prove destabilizing. In fact, in the aftermath of the January-February turmoil some senior bank executives complained that the rules for CoCos are too complicated and could undermine a bank's financial position rather than strengthen it in a crisis (Arnold and Hale, 2016). Concerns regarding these instruments were compounded by the uncertainties surrounding the interaction between the results of the EU wide stress test and the Pillar 2 requirement, together with the direct consequences of this interaction on the level of the MDA trigger point, i.e. the capital level below which a bank faces restrictions on the amount of distributable profits, including in the form of coupon payments on CoCos.

The ECB (European Central Bank, 2016b) clarified its position on July 29, 2016, in apparent coordination with the simultaneous publication of the stress test results by the EBA (European Banking Authority, 2016a). It announced that one element of the quantitative results of the stress test (that is the potential capital shortfall under adverse scenario) would be an input of the non-binding Pillar 2 guidance, a newly introduced element of the SREP process that, as opposed to Pillar 2 requirements, does not contribute to the determination of the MDA trigger. By doing so, the ECB not only reduced uncertainty on the supervisory framework in a dimension that is crucial for investors to understand the repayment behaviour of CoCos, but also de facto increased the protection of CoCos' investors by making it less likely that banks would face restrictions on the payment of coupons on AT1 instruments. In the European Central Bank (2016a) own words with "... the introduction of the component of Pillar 2 guidance, the capital requirements of a bank in terms of Pillar 1 plus Pillar 2 requirements will be reduced – all things being equal. As a result, the trigger for the Maximum Distributable Amount (MDA) will go down – also all things being equal...".

Both the reduction in uncertainty and the enhancement of protection are likely to have played a role in the weakening of the CoCo-specific contagion in the second DB stress period identified in our empirical analysis. In addition to that, the weakening could also result from a "learning process" from the part of investors spurred by the market disruption during the first stress period that made agents more aware of the riskiness of these instruments at the time of the second stress event, thereby rendering CoCos' prices less volatile in front of idiosyncratic shocks. More market stability in stressed periods is certainly reassuring regarding the convenience of CoCos as a substitute of costlier common equity for the purpose of increasing banks' loss absorption capacity.

We conclude the paper with a final comment. One of the most appraised features of CoCos is that of enabling the recapitalization of an institution as a going concern, that is, ahead of resolution. This property distinguishes CoCos from other AT2 instruments such as subordinated debt or the yet to be introduced bail-in eligible liabilities (MREL) that allow for the bank recapitalization as a gone concern, that is, in a resolution process. Yet, the supervisory actions described above went in the direction of relaxing the conditions that may lead to automatic restrictions on the payment of CoCos' coupons, thereby reducing the ability of these liabilities to provide for additional capital as a going concern, and rendering them de facto closer to AT2 instruments. While the ECB actions might have been justified (and the need for a reduction on uncertainty on supervisory treatment was uncontroversial), neither supervisory authorities nor investors should interpret them as an implicit recognition that "whatever it takes" will be done to stabilize CoCos markets. These instruments will only be useful if they achieve in practice what they are supposed to do by their design, namely the recapitalization of an institution as a going concern in a smooth manner that does not disrupt neither the institution itself nor the rest of the system.

References

- Admati, A.R., DeMarzo, P.M., Hellwig, M.F., Pfleiderer, P.C., 2013. Fallacies, irrelevant facts, and myths in the discussion of capital regulation: Why bank equity is not socially expensive. Max Planck Institute for Research on Collective Goods 2013/23.
- Allen, H.J., 2012. Cocos can drive markets cuckoo. Lewis & Clark L. Rev. 16, 125.
- Arnold, M., Hale, T., 2016. Ecb is having second thoughts on 'coco' bonds. Financial Times, April 24.
- Avdjiev, S., Bogdanova, B., Bolton, P., Jiang, W., Kartasheva, A., 2017. CoCo issuance and bank fragility. Technical Report. National Bureau of Economic Research.
- Avdjiev, S., Kartasheva, A.V., Bogdanova, B., 2013. Cocos: a primer. BIS Quarterly Review September.
- Basel Committee for Banking Supervision, 2011. Basel III: A global regulatory framework for more resilient banks and banking systems revised version June 2011. Bank for International Settlements.
- Bekaert, G., Ehrmann, M., Fratzscher, M., Mehl, A., 2014. The global crisis and equity market contagion. The Journal of Finance 69, 2597–2649.
- Bekaert, G., Harvey, C.R., Ng, A., 2005. Market integration and contagion. The Journal of Business 78, 39–69.
- Berg, T., Kaserer, C., 2015. Does contingent capital induce excessive risk-taking? Journal of Financial intermediation 24, 356–385.
- Blundell, R., Bond, S., 1998. Initial conditions and moment restrictions in dynamic panel data models. Journal of econometrics 87, 115–143.
- Calomiris, C.W., Herring, R.J., 2013. How to design a contingent convertible debt requirement that helps solve our too-big-to-fail problem. Journal of Applied Corporate Finance 25, 39–62.
- Calvo, G.A., Mendoza, E.G., 2000. Rational contagion and the globalization of securities markets. Journal of international economics 51, 79–113.
- Claessens, S., Dornbusch, R., Park, Y.C., 2001. Contagion: Why crises spread and how this can be stopped, in: International financial contagion. Springer, pp. 19–41.
- Davis, P.J., 2016. Deutsche bank to be hurt by mortgage penalty. The Wall Street Journal, September 19.

- Delivorias, A., 2016. Contingent convertible securities: is a storm brewing. European Parliamentary Research Service Briefing .
- Donahue, P., 2016. Merkel rules out assistance for deutsche bank. www.bloomberg.com/news/articles/2016-09-24/merkel-rules-out-assistance-fordeutsche-bank-focus-reports, September 24.
- Drukker, D.M., et al., 2003. Testing for serial correlation in linear panel-data models. Stata Journal 3, 168–177.
- Duffie, D., 2009. Contractual methods for out-of-court restructuring of systemically important financial institutions. Submission Requested by the US Treasury Working Group on Bank Capital, draft of November 9.
- European Banking Authority, 2015. Opinion of the European Banking Authority on the interaction of Pillar 1, Pillar 2 and combined buffer requirements and restrictions on distributions.
- European Banking Authority, 2016a. 2016 EU-wide stress test results.
- European Banking Authority, 2016b. EBA clarifies use of 2016 EU-wide stress test results in the SREP process.
- European Central Bank, 2016a. Frequently asked questions on the 2016 EU-wide stress test.
- European Central Bank, 2016b. SSM SREP Methodology Booklet 2015 edition.
- Federal Reserve Board, 2016. Comprehensive Capital Analysis and Review 2016: Assessment Framework and Results.
- Flannery, M.J., 2005. No pain, no gain? effecting market discipline via reverse convertible debentures. Capital adequacy beyond Basel: Banking, securities, and insurance, 171–196.
- Flannery, M.J., 2016. Stabilizing large financial institutions with contingent capital certificates. Quarterly Journal of Finance 6, 1650006.
- Forbes, K.J., 2012. The "big C": Identifying contagion. Technical Report. National Bureau of Economic Research.
- Forbes, K.J., Rigobon, R., 2002. No contagion, only interdependence: measuring stock market comovements. The Journal of Finance 57, 2223–2261.
- Glasserman, P., Nouri, B., 2012. Contingent capital with a capital-ratio trigger. Management Science 58, 1816–1833.
- Glover, J., 2016. Deutsche bank coco bonds have bumpy ride as lenders struggle.

www.bloomberg.com/news/articles/2016-01-28/deutsche-bank-coco-investors-have -bumpy-ride-as-lender-struggles, January 28.

- Goncharenko, R., Ongena, S., Rauf, A., 2017. The agency of coco: Why do banks issue contingent convertible bonds? .
- Goodhart, C., 2010. Are cocos from cloud cuckoo-land? Central Banking 21, 29–33.
- Hilscher, J., Raviv, A., 2014. Bank stability and market discipline: The effect of contingent capital on risk taking and default probability. Journal of Corporate Finance 29, 542–560.
- International Monetary Fund, 2016. Germany FSAP. IMF Country report no. 16/191. October.
- Martynova, N., Perotti, E., 2018. Convertible bonds and bank risk-taking. Journal of Financial Intermediation , https://doi.org/10.1016/j.jfi.2018.01.002.
- Pazarbasioglu, C., Zhou, M.J.P., Le Leslé, V., Moore, M., 2011. Contingent Capital: Economic Rationale and Design Features. IMF Staff Discussion Note SDN/11/01.
- Pennacchi, G., 2011. A structural model of contingent bank capital. Federal Reserve Bank of Cleveland Working Paper 10-04 .
- Pennacchi, G., Vermaelen, T., Wolff, C.C., 2014. Contingent capital: The case of coercs. Journal of Financial and Quantitative Analysis 49, 541–574.
- Pericoli, M., Sbracia, M., 2003. A primer on financial contagion. Journal of Economic Surveys 17, 571–608.
- Posner, K.A., 2010. Thoughts on the squam lake report: Reengineering the financial system to better withstand extreme volatility. Journal of Applied Corporate Finance 22, 34–39.
- Squam Lake Working Group, 2009. An expedited resolution mechanism for distressed financial firms: Regulatory hybrid securities. Council on Foreign Relations Working Paper 10.
- Stock, J.H., Watson, M.W., 2002. Macroeconomic forecasting using diffusion indexes. Journal of Business & Economic Statistics 20, 147–162.
- Strasburg, J., 2016. Deutsche faces plight of capital in u.s. litigation. The Wall Street Journal, September 19.
- Sundaresan, S., Wang, Z., 2015. On the design of contingent capital with a market trigger. The Journal of Finance 70, 881–920.
- Vallee, B., 2015. Contingent capital trigger effects: Evidence from liability management exercises. Unpublished working paper .
- Wooldridge, J.M., 2002. Econometric Analysis of Cross Section and Panel Data. The MIT Press.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1171 *The potential of big housing data: an application to the Italian real-estate market*, by Michele Loberto, Andrea Luciani and Marco Pangallo (April 2018).
- N. 1172 ECB monetary policy and the euro exchange rate, by Martina Cecioni (April 2018).
- N.1173 *Firms' investments during two crises*, by Antonio De Socio and Enrico Sette (April 2018).
- N.1174 How can the government spending multiplier be small at the zero lower bound?, by Valerio Ercolani and João Valle e Azevedo (April 2018).
- N. 1175 Asset price volatility in EU-6 economies: how large is the role played by the ECB?, by Alessio Ciarlone and Andrea Colabella (June 2018).
- N.1176 Fixed rate versus adjustable rate mortgages: evidence from euro area banks, by Ugo Albertazzi, Fulvia Fringuellotti and Steven Ongena (June 2018).
- N. 1177 Short term forecasts of economic activity: are fortnightly factors useful?, by Libero Monteforte and Valentina Raponi (June 2018).
- N. 1178 *Discretion and supplier selection in public procurement*, by Audinga Baltrunaite, Cristina Giorgiantonio, Sauro Mocetti and Tommaso Orlando (June 2018).
- N. 1179 *Labor market and financial shocks: a time varying analysis*, by Francesco Corsello and Valerio Nispi Landi (June 2018).
- N. 1180 On the unintended effects of public transfers: evidence from EU funding to Southern Italy, by Ilaria De Angelis, Guido de Blasio and Lucia Rizzica (June 2018).
- N. 1181 Always look on the bright side? Central counterparties and interbank markets during the financial crisis, by Massimiliano Affinito and Matteo Piazza (July 2018).
- N. 1182 *Knocking on parents' doors: regulation and intergenerational mobility*, by Sauro Mocetti, Giacomo Roma and Enrico Rubolino (July 2018).
- N. 1183 Why do banks securitise their assets? Bank-level evidence from over one hundred countries in the pre-crisis period, by Fabio Panetta and Alberto Franco Pozzolo.
- N. 1184 Capital controls spillovers, by Valerio Nispi Landi (July 2018).
- N.1185 The macroeconomic effects of an open-ended asset purchase programme, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (July 2018).
- N.1186 Fiscal buffers, private debt and recession: the good, the bad and the ugly, by Nicoletta Batini, Giovanni Melina and Stefania Villa (July 2018).
- N.1187 Competition and the pass-through of unconventional monetary policy: evidence from TLTROs, by Matteo Benetton and Davide Fantino (September 2018).
- N.1188 Raising aspirations and higher education: evidence from the UK's Widening Participation Policy, by Lucia Rizzica (September 2018).
- N. 1189 *Nearly exact Bayesian estimation of non-linear no-arbitrage term structure models*, by Marcello Pericoli and Marco Taboga (September 2018).
- N. 1190 *Granular Sources of the Italian business cycle*, by Nicolò Gnocato and Concetta Rondinelli (September 2018).
- N. 1191 *Debt restructuring with multiple bank relationships*, by Angelo Baglioni, Luca Colombo and Paola Rossi (September 2018).
- N.1192 Exchange rate pass-through into euro area inflation. An estimated structural model, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2018).
- N. 1193 *The effect of grants on university drop-out rates: evidence on the Italian case*, by Francesca Modena, Enrico Rettore and Giulia Martina Tanzi (September 2018).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **WP 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, WP 915 (June 2013).
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **WP 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **WP 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **WP 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **WP 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, *Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register,* Journal of Financial Intermediation, v. 25, 1, pp. 54-76, **WP 848 (February 2012).**
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, WP 1011 (June 2015).
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **WP 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, WP 965 (July 2015).
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, WP 932 (September 2013).
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **WP 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, WP 963 (June 2014).
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, WP 941 (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, WP 947 (January 2014).
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **WP 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **WP 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, WP 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, WP 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **WP 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **WP 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, WP 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **WP 1005 (March 2015).**
- GERALI A., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of simultaneous implementation of reforms*, International Finance, Wiley Blackwell, v. 19, 1, pp. 42-65, **WP 997 (December 2014).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, WP 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **WP 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, WP 1013 (June 2015).
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, WP 968 (July 2014).

2017

- AABERGE, R., F. BOURGUIGNON, A. BRANDOLINI, F. FERREIRA, J. GORNICK, J. HILLS, M. JÄNTTI, S. JENKINS, J. MICKLEWRIGHT, E. MARLIER, B. NOLAN, T. PIKETTY, W. RADERMACHER, T. SMEEDING, N. STERN, J. STIGLITZ, H. SUTHERLAND, *Tony Atkinson and his legacy*, Review of Income and Wealth, v. 63, 3, pp. 411-444, WP 1138 (September 2017).
- ACCETTURO A., M. BUGAMELLI and A. LAMORGESE, *Law enforcement and political participation: Italy 1861-65*, Journal of Economic Behavior & Organization, v. 140, pp. 224-245, WP 1124 (July 2017).
- ADAMOPOULOU A. and G.M. TANZI, *Academic dropout and the great recession*, Journal of Human Capital, V. 11, 1, pp. 35–71, **WP 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, *Information externalities in the credit market and the spell of credit rationing*, Journal of Financial Intermediation, v. 30, pp. 61–70, WP 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation,* Review of Economic Dynamics, v. 24, pp. 66-78, **WP 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, WP 761 (June 2010).
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, WP 1085 (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 WP 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, WP 1049 (January 2016).
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **WP 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, WP 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts*, Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **WP 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **WP 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, WP 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, WP 787 (January 2011).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **WP 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, WP 953 (March 2014).
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, WP 942 (November 2013).

- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, WP 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, *Adaptive models and heavy tails with an application to inflation forecasting*, International Journal of Forecasting, v. 33, 2, pp. 482-501, WP 1052 (March 2016).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **WP 877 (September 2012).**
- GIACOMELLI S. and C. MENON, Does weak contract enforcement affect firm size? Evidence from the neighbour's court, Journal of Economic Geography, v. 17, 6, pp. 1251-1282, WP 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **WP 1033 (October 2015).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **WP 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **WP 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, WP 752 (March 2010).
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **WP 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, WP 887 (October 2012).
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, WP 1073 (July 2016).
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, WP 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, WP 1048 (January 2016).
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **WP 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, WP 1065 (April 2016).
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, WP 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, WP 734 (November 2009).
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, WP 1104 (February 2017).

2018

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- BARONE G., G. DE BLASIO and S. MOCETTI, The real effects of credit crunch in the great recession: evidence from Italian provinces, Regional Science and Urban Economics, v. 70, pp. 352-59, WP 1057 (March 2016).
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, **WP 909 (April 2013).**

- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life,* Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME* borrowing. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

FORTHCOMING

- ACCETTURO A., W. DI GIACINTO, G. MICUCCI and M. PAGNINI, *Geography, productivity and trade: does* selection explain why some locations are more productive than others?, Journal of Regional Science, WP 910 (April 2013).
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **WP 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, WP 1098 (February 2017).
- BELOTTI F. and G. ILARDI, Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, WP 1147 (October 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, **WP 985 (November 2014).**
- COVA P., P. PAGANO and M. PISANI, Domestic and international macroeconomic effects of the Eurosystem Expanded Asset Purchase Programme, IMF Economic Review, WP 1036 (October 2015).
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, WP 787 (January 2011).

- D'IGNAZIO and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* Scandinavian Journal of Economics, **WP 900 (February 2013).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **WP 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **WP 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the atlantic,* Journal of Macroeconomics, WP 1140 (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, WP 898 (January 2013).
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, WP 1167 (March 2018).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, WP 1025 (July 2015).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, WP 871 (July 2012).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, WP 1100 (February 2017).